\begin{table}%t2
\caption{\label{fitresults}Spectral fit results for IBIS/ISGRI  data.} 
\par
\small%\centering
\par
\begin{tabular}{lrccccc}
\hline\hline
Name &  ISGRI & $N_{\rm H}$ & $f_{20-40~\rm keV}$ & $f_{40-100~\rm keV}$ & $\Gamma_{\rm ISGRI}^+$ & $\log L_{20 - 100~\rm keV}$\\
 &  18--60 keV $[\sigma]$ & $[10^{22}~\rm cm^{-2}]$ &  $[10^{-11}~\rm erg~cm^{-2}~s^{-1}]$ &  $[10^{-11}~\rm erg~cm^{-2}~s^{-1}]$  & &  $[\rm erg~s^{-1}]$ \\ 
\hline
IGR J00254+6822 & 7.5  & $40^c$   & 0.6 & 0.8 & $2.1 \pm 0.5$&
42.67  \\
IGR J00335+6126 & 5.9 & $0.5^*$ & 0.3 & 0.9 & $1.1 \pm 0.2$ & 44.50 \\
1ES 0033+59.5   & 12.6 & $0.36^a$  & 0.8 & 0.3 & $3.6 {+0.4 \atop
 -0.3}$ &44.36 \\   
Mrk 348         & 14.6 & $30^b$   & 4.2 & 6.6 & C   &43.74 \\ 
NGC 526A        & 4.3 & $1.6^b$ & 2.2 & 2.9 & 2 & 43.62 \\
ESO 297-18      & 5.2 & $42^i$  & 3.1 & 6.8 & $1.5 \pm 0.3$ &44.10 \\
IGR J01528-0326 & 8.5 & $14^c$  & 0.9 & 2.3 & $1.2 {+0.2 \atop -0.5}$ &
43.30\\
NGC 788         & 23.4 & ${<}0.02^a$ & 2.5 & 3.9 & $1.8 \pm 0.1$         & 43.42 \\
Mrk 590         & 4.5 & 0.03 & 0.5 & 0.6 & 2 & 43.23\\
IGR J02097+5222 & 7.7 & $0.03^i$ & 1.2 & 2.0 & $1.7 {+0.3 \atop -0.2}$ &44.25 \\
SWIFT J0216.3+5128 & 5.1 & $1.27^f$ & 0.8 & 0.8 & $2.3 {+1.2 \atop -1.0}$ &43.48 \\ 
Mrk 1040       & 3.4 & $0.067^b$ & 2.3 & 3.0 & 2 & 43.52 \\
IGR J02343+3229 & 3.5 & $2.2^d$ & 1.4 & 1.8 & 2 &                       43.27\\
NGC 985         & 5.8 & $0.6^b$ & 0.9 & 1.3 & $2.0 {+0.4 \atop -0.3}$ & 43.99 \\
NGC 1052        & 5.6 & $0.041^b$ & 0.8 & 1.5 & $1.6 {+0.8 \atop -0.4}$                    & 42.11 \\
RBS 345    &  5.0 & -- & 0.9 & 1.9 & $1.3 {+0.6 \atop -0.4}$ & 44.49 \\  
NGC 1068        &  7.3 & ${>}150^a$& 1.2 & 1.3 & $2.3 {+0.4 \atop -0.3}$ & 43.68\\
QSO B0241+62    & 16.3 & $1.5^a$ & 2.0 & 3.0 & $1.8 \pm 0.1$ & 44.37 \\
IGR J02466--4222 &  3.3 & $1$   & 1.0 & 1.4 & 2 & 44.45 \\
IGR J02501+5440 &  5.1 & --     & 0.6 & 1.5 & $1.2 \pm 0.3$ & 43.03 \\  
MCG-02-08-014   &  6.3 & --      & 1.3 & 0.9 & $2.7 {+0.6 \atop -0.5}$  & 43.15 \\ 
NGC 1142        & 16.2 & $45^b$  & 2.7 & 4.2 & $1.8  \pm 0.1$ & 44.11 \\ 
QSO B0309+411   &  4.0 & ${<}0.1^*$& 3.0 & 3.9 & 2 & 45.53 \\
NGC 1275        & 15.7 & $3.75^a$& 2.2 & 0.4 & $3.7 {+0.3 \atop -0.2}$ & 43.18\\
1H 0323+342     &  5.1 & $0.1^b$ & 0.7 & 2.4 & $1.0 {+0.4 \atop -0.7}$ &
44.43 \\ 
IGR J03334+3718 &  6.9 & --     & 1.0 & 1.5 & $1.9 \pm 0.3$ & 44.25\\
NGC 1365        &  5.1 & $44^b$  & 1.5 & 2.2 & $1.8 \pm 0.6$ & 42.38 \\
IGR J03532-6829 &  7.3 & $0.05^*$ & 1.4 & 0.6 & $3.5 {+0.7 \atop -0.6}$ & 44.63 \\ 
3C 111          & 13.7 & $0.63^a$& 5.3 & 6.9 & $2.0 \pm 0.2$ & 44.83 \\
3C 120          & 22.0 & $0.2^i$ & 3.0 & 4.4 & $1.8 \pm 0.1$ & 44.27 \\ 
UGC 3142        & 17.3 & $1.4^x$ & 2.8 & 3.7 & $2.0 \pm 0.1$ & 44.84 \\ 
LEDA 168563     &  5.4 & ${<}0.22^i$& 2.2 & 3.1 & $1.9 {+0.5 \atop -0.4}$&44.00 \\ 
ESO 33-2        & 15.5 & $0.1^b$ & 1.3 & 2.0 & $1.8 \pm 0.2$ & 43.38 \\
4U 0517+17      & 24.5 & $0.1^b$ & 3.0 & 3.6 & $2.13 {+0.09 \atop -0.08}$ &
43.68 \\
Ark 120         & 15.6 & ${<}0.1^*$ & 2.7 & 3.4 & $2.1 \pm 0.1$ & 44.18 \\
NGC 2110        &  4.9 & $4.3^*$& 7.0 & 9.3 & 2 & 43.34 \\
MCG+08-11-011   &  7.8 & $0.183^b$& 5.2 & 6.7 & $2.0 \pm 0.4$& 44.05 \\
IRAS 05589+2828 & 15.4 & ${<}0.04^i$& 1.4 & 2.8 & $1.5 {+0.2 \atop -0.1}$ &
44.02 \\
SWIFT J0601.9-8636 & 6.5 & $5.6^i$&1.3 & 1.0 & $2.6 {+0.7 \atop -0.6}$ &
42.32 \\
IGR J06117-6625 & 17.8 & $0.048^b$& 1.9 & 2.6 & $1.96 \pm 0.16$ & 45.85 \\
Mrk 3           & 26.7 & $110^a$  & 3.7 & 6.4 & $1.73 \pm 0.07$ & 43.61  \\
IGR J06239-6052 &  3.6 & $20^h$& 0.5 & 0.7 & 2 & 43.65 \\
PKS 0637-752    &  5.4 & $0.035^b$& 0.7 & 1.1 &  $1.7 \pm 0.6$ & 46.45\\
Mrk 6           & 15.1 & $10^a$& 1.9 & 2.3 & $2.1 \pm 0.2$ & 43.53 \\
QSO B0716+714   &  4.6 & ${<}0.01^a$& 0.3 & 0.3 & 2 & 45.24 \\
LEDA 96373      &  5.3 & -- & 1.4 & 2.6 & $1.5 {+0.7 \atop -0.5}$ & 43.89 \\
IGR J07437-5137&  3.5 & -- & 0.4 & 0.5 & 2 & 43.13 \\ 
IGR J07565-4139 &  6.2 & $1.1^b$& 0.7 & 0.6 & $2.7 {+0.6 \atop -0.4}$ & 43.12 \\
IGR J07597-3842 & 14.2 & $0.05^b$& 1.8 & 1.9 & $2.3 \pm 0.2$ & 44.14 \\
ESO 209-12      & 12.2 & $0.1^b$& 0.9 & 1.4 & $1.8 {+0.1 \atop -0.2}$ & 43.93 \\
PG 0804+761     &  6.3 & $0.023^b$ & 1.1 & 1.5 & $2.0 \pm 0.5$ & 44.81\\
Fairall 1146    &  9.9 & $0.1^b$& 0.9 & 1.4 & $1.8 {+0.3 \atop -0.2}$&43.73\\
QSO B0836+710   & 15.4 & $0.11^a$& 2.1 & 4.1 & $1.5 {+0.2 \atop -0.1}$ & 47.93\\
IGR J09026-4812  & 17.8 & $0.9^w$ & 0.9 & 1.4 & $1.9 \pm 0.1$ & 43.91\\
SWIFT J0917.2-6221&5.8 & $0.5^f$& 0.8 & 1.1 & $1.9 {+0.5 \atop -0.6}$   & 44.18 \\
IGR J09253+6929 &  7.0 & $8^d$& 1.8 & 2.6 & $1.9 \pm 0.4$ & 44.19\\
Mrk 110        &  3.6 & $0.019^b$& 2.2 & 3.0 & 2 & 44.18 \\
IGR J09446-2636&  3.2 & $0.1^b$& 1.3 & 1.7 & 2 & 45.60 \\
\hline
NGC 2992        & 15.8 & $0.1^l$& 2.8 & 4.1 & $1.9 \pm 0.1$ & 42.95\\
MCG-05-23-016   & 14.6 & $1.6^a$ & 5.6 & 8.1 & $1.9 \pm 0.1$ & 43.34\\
IGR J09523-6231 &  5.3 & $8^d$ & 0.6 & 0.7 & $2.2 {+0.6 \atop -0.5}$ & 45.41 \\ 
NGC 3081        &  9.7 & $66^m$ & 2.6 & 4.1 & $1.8 \pm 0.2$ & 42.97 \\
SWIFT J1009.3-4250&6.9 &  $30^f$ & 1.5 & 2.0 & $2.0 \pm 0.4$ & 43.95\\
IGR J10147-6354&  3.0 & $2^d$  & 0.5 & 0.7 & 2 & 45.18\\
NGC 3227        & 16.6 & $6.8^b$ & 5.6 & 7.1 & $2.0 \pm 0.1$ & 42.63\\
NGC 3281        &  7.5 & $151^b$ & 1.8 & 2.7 & $1.9 \pm 0.3$ & 43.06\\
SWIFT J1038.8-4942&5.3 & $0.6^f$ & 0.6 & 1.4 & $1.3 \pm 0.4$ & 44.21 \\
IGR J10404-4625 &  6.9 & $3^f$ & 1.2 & 1.4 & $2.2 {+0.4 \atop -0.3}$ & 43.54\\
Mrk 421         &173.8 & $0.08^i$ & 22.4 & 20.6 & $2.45 {0.03 \atop -0.02}$ & 44.92\\
IGR J11366-6002 &  7.6 & $0.35^o$ & 0.6 & 0.7 & $2.2 {+0.5 \atop -0.4}$ & 42.73 \\
NGC 3783       &  6.1 & $0.08^*$ & 6.0 & 8.4 & $1.9 \pm 0.4$ & 43.48\\
IGR J12026-5349 & 14.8 & $2.2^a$ & 1.6 & 2.2 & $2.0 \pm 0.1$ & 43.84\\
NGC 4051        & 11.7 &  ${<}0.02^*$ & 1.8 & 1.5 & $2.1 \pm 0.2$ & 41.58 \\
NGC 4138        &  9.0 & $8^b$ & 1.2 & 1.9 & $1.8 \pm 0.3$ & 41.79 \\
NGC 4151       & 205.7 & $6.9^a$ & 24.0 & 32.3 & C & 43.13\\
NGC 4180        &  9.4 & -- & 0.9 & 1.8 & $1.6 {+0.3 \atop -0.2}$ & 42.46 \\
Was 49         &  3.8 & $10^b$ & 0.6 & 0.8 & 2 & 44.12\\
Mrk 766         &  6.9 & $0.8^a$ & 1.0 & 1.5 & $1.8 \pm 0.3$ & 42.95\\
NGC 4258        &  6.6 & $8.7^b$ & 1.1 & 1.1 & $2.3 \pm 0.4$ & 41.04\\
4C 04.42        &  7.7 & $0.1^b$  & 0.7 & 1.8 & $1.2 \pm 0.2$ & 46.83\\
Mrk 50          &  6.2 & $0.018^b$ & 0.5& 0.4 & $2.6 {+0.7 \atop -0.6}$ & 43.09\\
NGC 4388        & 78.1 & $27^a$ & 9.8 & 15.2 & C & 43.59\\
NGC 4395        &  8.8 & $0.15^a$ & 1.5 & 1.6 & $2.3 {+0.4 \atop -0.3}$ & 40.85\\
3C 273          & 78.0 & $0.5^a$ & 7.5 & 10.6 & $1.92 \pm 0.03$ & 46.09\\
NGC 4507        & 30.9 & $29^a$ & 6.3 &10.0 & C & 43.70\\
SWIFT J1238.9-2720&6.3 & $60^f$ & 2.7 & 5.3 & $1.6 {+0.4 \atop -0.3}$ & 44.06\\
IGR J12391-1612 &  9.9 & $3^f$ & 1.7 & 2.2 & $2.0 \pm 0.2$ & 44.08\\
NGC 4593        & 33.0 & $0.02^a$ & 3.2 & 4.1 & C & 43.08\\
IGR J12415-5750 &  6.7 & ${<}0.11^b$ & 0.8 & 1.4 & $1.7 {+0.2 \atop -0.3}$ & 43.45\\
PKS 1241-399    &  3.5 & -- & 0.7 & 0.9 & 2 & 45.23\\
ESO 323-32      &  5.6 & $7^f$ & 0.8 & 1.5 & $1.6 {+0.4 \atop -0.3}$ & 43.13\\
3C 279          &  8.6 & ${\la}0.13^a$ & 0.9 & 1.7 & $1.6 \pm 0.2$ & 46.40\\
Mrk 783        &  4.7 & $0.046^b$ & 0.9 & 1.2 & 2  & 44.36\\
IGR J13038+5348 &  4.5 & ${<}0.03^*$ & 1.0 & 1.3 & 2 & 43.67\\
NGC 4945        & 78.7 & $400^a$ & 9.9 & 18.4 & C & 42.35\\
IGR J13057+2036 &  3.1 & --  & 0.4 & 0.6 &  2   & \\
ESO 323-77      &  9.0 & $55^a$ & 1.1 & 2.1 & $1.5 {+0.3 \atop -0.2}$ & 43.21\\
IGR J13091+1137 &  7.5 & $90^a$ & 1.6 & 2.5 & $2.1 \pm 0.3$           & 43.76\\
IGR J13109-5552 &  8.7 & ${<}0.1^p$  & 0.9 & 1.2 & $2.0 {+0.3 \atop -0.2}$ & 44.78\\
NGC 5033        &  4.2 & $0.03^b$ & 0.6 & 0.8 & 2  & 41.43\\
IGR J13149+4422 & 4.7 & $5^d$ & 0.8 & 1.1 & 2 & 43.77\\
Cen A           &247.2 & $12.5^a$ & 28.1 & 43.4 & $1.82 \pm 0.01$ & 42.71\\
ESO 383-18      &  6.5 & $17^x$ & 0.9 & 1.2 & $1.9 {+0.4 \atop -0.3}$ & 42.86\\
MCG-06-30-015   & 21.1 & $0.03^b$ & 2.3 & 1.7 & $2.4 \pm 0.1$ & 42.74\\
NGC 5252        &  3.0 & $0.68^b$  & 2.0 & 2.7 & 2 & 43.76\\
Mrk 268         &  5.7 & --  & 0.9 & 0.7 & $2.5 \pm 0.6$ & 43.78\\
4U 1344-60      & 35.2 & $5^a$ & 3.5 & 4.6 & C & 43.48\\
IC 4329A        & 58.5 & $0.42^a$ & 8.8 & 12.3 & C & 44.09\\
Circinus Galaxy &120.0 & $360^a$  & 10.9 & 10.1 & C & 41.96\\
NGC 5506        & 27.8 & $3.4^a$  & 7.0 & 6.2 & $2.1 \pm 0.1$ & 43.16\\
IGR J14175-4641 &  8.0 & --       & 0.8 & 1.3 & $1.8 \pm 0.3$& 44.48\\
NGC 5548        &  3.1 & $0.51^a$  & 1.0 & 1.3 & 2 & 43.19\\
RHS 39          &  7.7 & ${<}0.05^a$ & 1.6 & 2.2 & $2.0 \pm 0.3$& 43.63\\
H 1426+428      &  4.6 & ${<}0.02^*$ & 0.6 & 0.8 & 2 & 44.80\\
IGR J14471-6414 &  7.2 & ${<}0.1^*$ & 0.5 & 0.8 & $1.8 \pm 0.3$    & 43.95\\
IGR J14471-6319 &  6.4 & $2^f$ & 0.6 & 0.8 & $2.0 {+0.5 \atop -0.6}$    & 43.67\\
IGR J14492-5535 &  8.6 & $12^q$ & 0.9 & 1.4 & $1.7 \pm 0.3$ & \\
\hline
IGR J14515-5542 &  9.4 & $0.4^f$ & 0.8 & 1.4 & $1.6 \pm 0.2$  & 43.19\\
IGR J14552-5133 &  8.5 & $0.1^b$ & 0.6 & 1.1 & $1.6 {+0.4 \atop -0.2}$  & 42.98\\
IGR J14561-3738 &  9.4 & ${>}100^q$ & 0.6 & 1.2 & $1.7 {+0.2 \atop -0.4}$ & 43.37\\
IGR J14579-4308 & 15.4 & $20^d$  & 1.1 & 1.4 & $2.0 \pm 0.2$  & 43.16\\
Mrk 841         &  3.2 & $0.21$ & 1.0 & 1.4 & 2 & 43.87\\
ESO 328-36      &  5.7 & -- & 0.4 & 0.4 & $2.5 \pm 0.6$            & 43.00\\
IGR J15161-3827 & 12.3 & -- & 1.0 & 1.7 & $1.7 \pm 0.2$            & 43.90\\
NGC 5995        & 16.4 & $0.7^*$ & 2.2 & 1.5 & $2.8 \pm 0.2$    & 43.74\\
IGR J15539-6142 & 4.6 & $18^f$ & 0.5 & 0.7 & 2 & 42.78\\
IGR J16024-6107 &  3.7 & ${<}0.1^o$ & 0.3 & 0.4 & 2             & 42.46\\
IGR J16056-6110 &  3.8 & ${<}1^c$ & 0.3 & 0.5 & 2 & 43.72\\
IGR J16119-6036 &  9.3 & $0.1^b$ & 0.9 & 1.4 & $1.8 {+0.3 \atop -0.2}$ & 43.11\\
IGR J16185-5928 &  6.1 & ${<}0.1^e$ & 0.5  & 0.9 & $1.6 \pm 0.3$  & 43.57 \\
IGR J16351-5806 & 10.2 & ${<}0.1^s$ & 0.8 & 1.6 & $1.5 \pm 0.2$ & 42.64\\
IGR J16385-2057 &  7.2 & $0.21^d$ & 0.8 & 0.5 & $3.1 \pm 0.4$           & 43.35\\
IGR J16426+6536 &  4.4 &    --    & 2.6 & 3.4 & 2             & 46.32\\
IGR J16482-3036 & 27.5 & $0.13^f$ & 2.5 & 3.1 & $2.1 \pm 0.1$  & 44.10\\
ESO 138-1       &  3.7 & $150^b$ & 0.4 & 0.6 & 2 & 42.26\\
NGC 6221       &  4.7 & $1^a$ & 0.5 & 1.0 & 2 & 41.93\\
NGC 6240        & 12.1 & $2^b$ & 2.1 & 3.6 & $1.7 \pm 0.2$          & 43.89\\
Mrk 501         & 12.4 & $0.013^b$ & 2.5 & 1.7 & $2.8 \pm 0.3$& 44.05\\
IGR J16558-5203 & 16.5 & $0.011^b$ & 1.1 & 1.5 & $2.0 \pm 0.1$     & 44.25\\
IGR J16562-3301 &  7.0 & $0.2^f$ & 1.0 & 2.3 & $1.32 {+0.16 \atop -0.07}$ & \\
NGC 6300        & 10.1 & $22^a$ & 3.1 & 3.6 & $2.2 \pm 0.2$ & 42.30\\
IGR J17204-3554 &  6.4 & $12^b$ & 0.3 & 0.4 & $2.2 \pm 0.5$ & \\
GRS 1734-292    &106.4 & $3.7^a$ & 4.3 & 4.4 & C         & 43.96\\
IGR J17418-1212 &  6.6 & $0.1^b$  & 0.8 & 1.5 & $1.5 \pm 0.2$ & 43.85\\
IGR J17488-3253 & 27.5 & $0.7^*$ & 1.1 & 1.6 & C       & 43.39\\
IGR J17513-2011 & 16.0 & $0.6^*$ & 0.8 & 1.1 & $1.95 \pm 0.12$ & 44.00  \\ 
IGR J18027-1455 & 15.2 & $19.0^a$ & 1.2 & 1.7 & $1.9 \pm 0.1$ & 41.86\\
IGR J18244-5622 &  3.0 & $14^f$ & 1.3 & 1.7 & 2                    & 43.28\\
IGR J18249-3243 &  7.5 & ${<}0.1^n$ & 0.5 & 0.7 & $1.9 \pm 0.4$ & 45.72\\
IGR J18259-0706 &  7.8 & $0.6^f$ & 0.6 & 0.8 & $2.1 \pm 0.3$ & \\
PKS 1830-211    & 25.2 & ${\la}0.7^a$ & 1.9 & 3.7 & $1.49 {+0.05 \atop
 -0.07}$ & 48.19\\ 
3C 382          &  3.0 & $0.88^b$ & 3.8 & 5.0 & 2 & 44.85\\
ESO 103-35     &  4.9 & $19^a$ & 3.0 & 4.0 & 2  & 43.44\\
3C 390.3        & 14.3 & ${<}0.1^a$ & 2.3 & 3.9& $1.7 \pm 0.1$  & 44.67\\
ESO 140-43      & 14.0 & $1.8^x$ & 1.7 & 2.6 & $1.7 \pm 0.2$ & 43.40\\
IGR J18559+1535 & 12.0 & $0.7^d$ & 0.9 & 1.3 & $2.0 \pm 0.2$  & 44.59\\
ESO 141-55      & 19.0 & $0.004^t$ & 1.9 & 2.8 & $1.9 \pm 0.4$ & 44.30\\
1RXS J192450.8-29143& 4.6 & $0.088^b$ & 0.5 & 0.7 & 2 & 45.70\\
1H 1934-063     &  9.2 & $0.1^b$ & 1.3 & 0.7 & $3.1 {+0.4 \atop -0.3}$  & 42.71\\
IGR J19405-3016 & 11.8 & ${<}0.1^o$ & 1.3 & 1.9 & $1.8 \pm 0.2$ & 44.30\\
NGC 6814        & 16.2 & ${<}0.05^a$ & 3.0 & 4.6 & $1.8 \pm 0.1 $  & 42.65\\
IGR J19473+4452 &  7.1 & $11^a$ & 1.1 & 2.0 & $1.3 \pm 0.3$   & 44.33\\
3C 403          &  4.5 & $45^b$ & 0.9 & 1.2 & 2   & 44.23\\
QSO B1957+405   & 42.6 & $20^a$ & 3.7 & 5.0 & $1.97 \pm 0.05$   & 44.81\\
ESO 399-20     &  4.3 & $0.048^b$ & 0.6 & 0.9 & 2 & 43.33\\
IGR J20187+4041 & 12.2 & $6.1^b$ & 0.9 & 1.3 & $1.9 {+0.1 \atop -0.2}$ & 42.99\\
IGR J20286+2544 & 10.4 & $42^f$ & 1.8 & 3.3 & $1.6 \pm 0.2$  & 43.28\\
Mrk 509         & 11.2 & ${<}0.01^a$ & 3.8 & 4.1 & $2.3 \pm 0.3$& 44.34\\
S5 2116+81     &  4.4 & ${<}0.1^a$ & 1.3 & 1.8 & 2 & 44.76\\
IGR J21178+5139 &  7.7 & $2^f$ & 0.8 & 1.5 & $1.5 \pm 0.2$    & \\
IGR J21247+5058 & 47.3 & $0.6^k$ & 4.7 & 7.3 & C & 44.04\\
IGR J21277+5656 & 13.6 & $0.1^b$ & 1.6 & 1.8 & $2.2 \pm 0.2$            & 43.20\\
RX J2135.9+4728 &  9.2 & $0.4^i$ & 0.9 & 1.6 & $1.7 \pm 0.2$            & 43.55\\
PKS 2149-306    &  4.6 & 0.03 & 0.9 & 1.2 & 2 & 47.98\\
NGC 7172        & 19.4 & $9.0^a$ & 3.6 & 4.7 & $2.0 \pm 0.1$   & 43.13\\
BL Lac          &  5.5 & $0.3^b$ & 0.7 & 1.2 & $1.8 {+0.4 \atop -0.3}$ & 44.34\\[2pt]
\hline
IGR J22292+6647 &  5.7 & $0.2^*$ & 0.5 & 0.6 & $2.3 \pm 0.5$ & 44.57\\
NGC 7314        & 12.9 & $0.122^b$ & 2.1 & 2.5 & $2.2 \pm 0.3$ & 42.37 \\
Mrk 915         &  4.5 & ${<}0.1^*$ & 0.6 & 0.8 & 2    & 43.28\\
IGR J22517+2217 &  6.6 & $3^v$ & 1.6 & 3.4 & $1.4 \pm 0.4$ & 48.43\\
3C 454.3        & 30.8 & $0.5^b$ & 7.1 & 13.2 & $1.58 \pm 0.06$ &  47.76\\
1H 2251-179     & 22.9 & ${<}0.19^a$ & 3.3 & 3.8 & $2.2 \pm 0.1$  & 44.85\\
NGC 7469        &  6.1 & $0.061^b$ & 1.8 & 2.3 & $2.1 \pm 0.4$  & 44.39\\
MCG-02-58-022   & 15.0 & ${<}0.08^a$ & 2.0 & 1.4 & $2.8 \pm 0.2$   & 44.26\\
IGR J23206+6431 &  7.7 & $0.6^*$ & 0.4 & 0.7 & $1.5 {+0.5 \atop -0.3}$  & 44.15\\
IGR J23308+7120 &  3.7 & $6^o$ & 0.3 & 0.4 & 2 & 43.39\\
IGR J23524+5842 &  6.7 & $6^d$ & 0.4 & 0.5 & $2.0 \pm 0.5$ & 44.79\\
\hline
\end{tabular}
\medskip
$^+$ ``C'' indicates that a more complex model is required to fit the data (see Table~\ref{complexISGRIspec}); $^*$~this work; $^a$ Beckmann et~al. (2006), and references therein; $^b$~Bodaghee et~al. (\cite{Bodaghee07}), and ref. therein; $^c$~Landi et~al. (\cite{Landi07b}); $^d$~Rodriguez et~al. (\cite{Rodriguez08}); $^e$~Malizia et~al. (\cite{Malizia08}); $^f$~Malizia et~al. (\cite{Malizia07}); $^g$~Beckmann et~al. (\cite{Beckmann07b}); $^h$~Revnivtsev et~al. (\cite{Revnivtsev07}); $^i$~Winter et~al. (\cite{Winter08}); $^k$~Ricci et~al. (\cite{Ricci09a}); $^l$~Beckmann et~al. (\cite{Beckmann07a}); $^m$~Bassani et~al. (\cite{Bassani99}); $^n$~Landi et~al. (\cite{Landi08}); $^o$~Landi et~al. (\cite{Landi07c}); $^p$~Molina et~al. (\cite{Molina08}); $^q$~Sazonov
 et~al. (\cite{Sazonov08}); $^s$~Landi et~al. (\cite{Landi07a}); $^t$~Gondoin et~al. (\cite{Gondoin03}); $^u$~Landi et~al. (\cite{Landi07d}); $^v$~Bassani et~al. (\cite{Bassani07}); $^w$~Tomsick et~al. (\cite{Tomsick08}); $^x$~Ricci et~al. (\cite{Ricci09b}).
\end{table}